Author’s response to reviews

Title: The role of collective affective commitment in the relationship between work-family conflict and emotional exhaustion among nurses: A multilevel modeling approach

Authors:

Maura Galletta (maura.galletta@gmail.com)
Igor Portoghese (igor.portoghese@gmail.com)
Paola Melis (pirimpola@hotmail.com)
Cesar Ivan Aviles Gonzalez (infermiere2010@gmail.com)
Gabriele Finco (gabriele.finco@gmail.com)
Ernesto D’Aloja (ernestodaloja@gmail.com)
Paolo Contu (contumail@gmail.com)
Marcello Campagna (mam.campagna@gmail.com)

Version: 2 Date: 29 Jan 2019

Author’s response to reviews:

Dear Reviewer,

Thank you very much for your appreciation about our revision and your last interesting comments.

Concerning the measures we used, those have previously been published and we provided the references in the manuscript.

Thank you very much for your important suggestions in improving the quality of our manuscript.